Author’s response to reviews

Title: A Chinese patient with 11β-hydroxylase deficiency due to novel compound heterozygous mutation in CYP11B1 gene: a case report

Authors:

Xianxian Yuan (xianxianyuan@163.com)
Lin Lu (lulin88@sina.com)
Shi Chen (cs0083@126.com)
Jun Jiang (jiangjun02009@126.com)
Xiangqing Wang (wxq521643@163.com)
Zhihui Liu (123284390@qq.com)
Huijuan Zhu (shengxin2004@163.com)
Hui Pan (panhui20111111@163.com)
Zhaolin Lu (luzhaolin39@163.com)

Version: 3 Date: 03 Sep 2018

Author’s response to reviews:

Dear Editor,

I am very grateful to your comments for the manuscript. According with your advice, we amended the relevant part in manuscript. Some of your questions were answered below.

1) Please have the text edited by a professional language editing service or a native English speaking colleague. There are many issues with grammar, wording, spelling, and/or punctuation that need to be addressed before acceptance for publication.

We have our manuscript reviewed by LetPub and we can provide the signed consent for at any stage if you request.
2) At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

We have uploaded the revised manuscript, as well as all relevant tables/figures/additional files, as a single, final, clean version.

3) Supplementary Table S1: please provide a short summary title in the main Word file.

We have provided a short summary title ‘Primers Used for PCR Assay of CYP11B1 Gene’ in the Genetic analysis section, line 10, page 6.

Thank you and all the reviewers for the kind advice.

Sincerely yours,

Xianxian Yuan

Corresponding author: Lin Lu; E-mail: lulin88@sina.com